Author's response to reviews

Title: Pathways to the diagnosis of lung cancer in the UK: a cohort study.

Authors:

   Jacqueline Barrett Ms (j.barrett@caperresearch.org)
   William T Hamilton Dr (w.hamilton@bristol.ac.uk)

Version: 4  Date: 12 May 2008

Author's response to reviews:

I've certainly sorted the author's list. Text now all black. I'm not sure the cropped figure is much better - in terms of space saving that is.